Author's response to reviews

Title: OPA1-related dominant optic atrophy is not strongly influenced by mitochondrial DNA background

Authors:

Denis Pierron (d.pierron@anthropologie.u-bordeaux1.fr)
Marc Férré (MaFerre@chu-angers.fr)
Christophe Rocher (crocher@u-bordeaux2.fr)
Arnaud Chevrolier (archevrollier@chu-angers.fr)
Pascal Murail (p.murail@anthropologie.u-bordeaux1.fr)
Didier Thoraval (didier.thoraval@lbms.u-bordeaux2.fr)
Patrizia Amati-Bonneau (PaBonneau@chu-angers.fr)
Pascal Reynier (PaReynier@chu-angers.fr)
Thierry Letellier (thierry.letellier@phys-mito.u-bordeaux2.fr)

Version: 3 Date: 31 May 2009

Author's response to reviews: see over
Dear BioMed Central Editorial Team,

We have carefully considered yours comments and corrected our manuscript.

Experimental research is reported in the manuscript have been performed with the approval of the ethics committee of “centre hospitalier universitaire d’Angers”. And the research was carried out on humans must be in compliance with the Helsinki Declaration. A statement to this effect have been added in the Methods section of the manuscript, including the name of the body which gave approval:

METHOD SECTION p.4 :

“Genetic analyses were carried out with the appropriate consent of the patients and in compliance with the Helsinki Declaration. Experimental research have been performed with the ethics approval of the “centre hospitalier universitaire d’Angers” and ethics policy of the French mitochondrial disease network [9]. “

Please don't hesitate to contact me if you have any problems or questions regarding us manuscript.

Yours sincerely,

Thierry Letellier
Directeur de l’Unité INSERM 688